Author's response to reviews

Title: Review: Ischemic Renal Injury

Authors:

Raj Munshi (Raj.Munshi@seattlechildrens.org)
Christine Hsu (cwhsu8@uw.edu)
Jonathan Himmelfarb (himmej@u.washington.edu)

Version: 4 Date: 18 November 2010

Author's response to reviews:

To the Editor and Reviewers,

Thank you for your thoughtful comments. In addressing your remarks, we believe we have improved upon the original draft. We have tracked our changes in red and have inserted them within the body of the text. We hope that you find the changes addressed your questions and look forward to any further observations you may have. In regards to the figures: The figures are our own creation. The labeling in figure 1 is correct. Figure 1A shows the passive stretch that occurs due to increased pressure, and 1C shows the timing and duration of the three regulatory mechanisms. We have also placed the references with the figure and not with the main text. We can place them with the main text if preferable.

Thank you for your time and the opportunity to discuss this interesting topic.

Sincerely,

Raj Munshi, Christine Hsu and Jonathan Himmelfarb.